Author's response to reviews

Title: Neutrophil/Lymphocyte Ratio has no predictive or prognostic value in breast cancer patients undergoing preoperative systemic therapy

Authors:

Christoph Suppan (christoph.suppan@klinikum-graz.at)
Vesna Bjelic-Radisic (vesna.bjelic-radisic@medunigraz.at)
Marlen La Garde (marlen.lagarde@stud.medunigraz.at)
Andrea Groselj-Strele (andrea.strele@klinikum-graz.at)
Katharina Eberhard (katharina.eberhard@medunigraz.at)
Hellmut Samonigg (hellmut.samonigg@medunigraz.at)
Hans Loibner (hans.loibner@apeiron-biologics.com)
Nadia Dandachi (nadia.dandachi@medunigraz.at)
Marija Balic (marija.balic@medunigraz.at)

Version: 4 Date: 13 November 2014

Author's response to reviews:

The requested comments have been taken into consideration and the following changes have been made:

1. Line and page numbers have been included.
2. The section „Acknowledgements“ has been included on page 11 and from our point of view there are no specific acknowledgements to anyone who does not meet the criteria for authorship.